Author’s response to reviews

Title: Respiratory compromise in children presenting to an urban emergency department of a tertiary hospital in Tanzania: A descriptive cohort study

Authors:

Biita Muhanuzi (biitamuhanuzi@yahoo.com)

Hendry Sawe (hendry_sawe@yahoo.com)

Said Kilindimo (skilindimo@yahoo.com)

Juma Mfinanga (jumamfinanga@gmail.com)

Ellen Weber (ellen.weber@me.com)

Version: 1 Date: 07 Aug 2018

Author’s response to reviews:

Hendry R. Sawe
Emergency Medicine Department
MUHAS
P.O. Box 65001
Dar es salaam Tanzania

7th August 2018

Editor in Charge,
BMC-Emergency Medicine

Dear Editor,

Re: Re-Submission of manuscript titled “Respiratory compromise in children presenting to an urban emergency medicine of a tertiary hospital in Tanzania: A descriptive cohort study”
Kindly refer to the above title.

I have addressed all the comments provided in the e-mail.

We hope you will consider our manuscript for publication in your journal.

Thank you in advance

Sincerely

Hendry R. Sawe